Author’s response to reviews

Title: Peri-conceptional or pregnancy exposure of HPV vaccination and the risk of spontaneous abortion: a systematic review and meta-analysis

Authors:

Jing Tan (tanjing84@outlook.com)
Yi-quan Xiong (xiongyq2018@126.com)
Qiao He (15802498951@163.com)
Yan-mei Liu (lymsmile@126.com)
Wen Wang (wangwen83@outlook.com)
Meng Chen (526543200@qq.com)
Kang Zou (zoukang528@hotmail.com)
Xing-hui Liu (xinghuiliu@163.com)
Xin Sun (sunxin@wchscu.cn)

Version: 2 Date: 21 Jul 2019

Author’s response to reviews:

Response to comments

-- Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Thanks for the comment. We have revised the authors' names using full initials in the Authors’ Contributions section (P18, L18-24)

-- Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Thanks for the comment. The statement: "All authors have read and approved the final version of the manuscript" has been added in the Authors’ Contributions section (P18, L25) .
In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Thanks for the comment. The statement: "The funders had no involvement in the study design, data collection and analysis, decision to publish or preparation of the manuscript."

" has been added in the Funding section (P18, L15-16).

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Thanks for the comment. The manuscript has been upload according to the above requirements.